food & nutrition (

ORIGINAL ARTICLE

Body mass index variation in adults with Williams syndrome: associations with predicted dietary intake and food behaviors

Danielle Renzi¹, Takara Stanley^{2,3}, Jessica Waxler¹, Hang Lee^{3,4}, Barbara Pober^{1,3*†} and Marianne Nordstrom^{5,6,7†}

¹Division of Genetics, Department of Pediatrics, Massachusetts General Hospital, Boston, MA, USA; ²Metabolism Unit, Department of Medicine, and Pediatric Endocrine Unit, Department of Pediatrics, Massachusetts General Hospital, Boston, MA, USA; ³Harvard Medical School, Boston, MA, USA; ⁴Biostatistics Center, Massachusetts General Hospital, Boston, MA, USA; ⁵Department of Nutrition, Institute of Basic Medical Sciences, University of Oslo, Oslo, Norway; ⁶Frambu Resource Centre for Rare Disorders, Siggerud, Norway; ⁷Unit for Rare Neuromuscular Disorders, Movement, Muscle and Neurodegeneration, Department of Neurology, Oslo University Hospital, Oslo, Norway

Popular scientific summary

- Variability in BMI and patterns of weight change over time were observed in a group of adults with Williams syndrome, a rare genetic disorder associated with intellectual disability.
- Dietary intakes of individuals with Williams syndrome deviated from national recommendations; however, few of the dietary intake components were related to BMI variability.
- Selected food-related problem behaviors were found to be associated with increased BMI or higher rate of weight gain.

Abstract

Background: Dietary intake and body weight are important predictors of long-term health. However, few studies have focused on these topics in adults with genetic syndromes that have associated intellectual disability, such as Williams syndrome (WS).

Objective: In adults with WS, describe predicted dietary intake, food-related problems, and associations between body mass index (BMI) and possible factors contributing to differences in weight status.

Design: In this study of 82 participants (median age of 30 years, range 18–69), we cross sectionally investigated associations between BMI, predicted dietary intakes (Dietary Screener Questionnaire), food-related behaviors (Food-Related Problem Questionnaire), and anxiety (Spence Children's Anxiety Scale). Longitudinal patterns of weight change were further studied in a subset (n = 41).

Results: BMI variation was observed with median BMI of 27.3 kg/m² (range 16.7–55.5 kg/m²). Several components of dietary intake deviated from recommendations in the WS cohort. When compared with WS participants with either normal or overweight BMI, WS participants with obesity had reduced daily intake of fruits and vegetables of 0.15 cup equivalents (P = 0.049), while participants with underweight BMI had reduced daily intake of fruits and vegetables of 0.44 cup equivalents (P = 0.026) and additionally had reduced intake of dietary fiber of 2.12 grams per day (P = 0.019). A one-point increase in the 'preoccupation with food' sub-score was associated with a 0.57 unit increase in BMI (P = 0.16), while a one-point increase in the 'takes and stores food' sub-score was associated with a 0.72 unit increase in BMI. In the longitudinal weight subset, a weight gain group and a weight stable group were identified. The former was associated with increase 'takes and stores food' sub-score but not with dietary intakes.

Conclusion: We observed considerable BMI variability. While few dietary intakes were associated with BMI, increased BMI and weight gain were associated with 'preoccupation with food' and with 'takes and stores food' behavior sub-scores.

[†]These authors contributed equally to this work.

Keywords: Williams syndrome; body weight; food frequency questionnaire; food behavior; weight trajectory

To access the supplementary material, please visit the article landing page

Received: 18 January 2023; Revised: 23 May 2023; Accepted: 24 May 2023; Published: 23 June 2023

ealthy diet and body weight are important for long-term health and well-being. An abundance of information documents the negative health effects of having a body mass index (BMI) ≥30 kg/m² (classified as obesity) or a BMI < 18.5 kg/m^2 (classified as being underweight). Less scientific attention has been paid to the issue of weight status among adults with intellectual and developmental disability (IDD), despite studies demonstrating an increased frequency and earlier onset of obesity compared with the general population (1-3). Here we present work on one specific multisystem genetic disorder associated with IDD, Williams syndrome (WS). WS is caused by a chromosome 7q11.23 microdeletion, which produces a loss of one copy of 26-28 unique sequence genes, including the elastin gene. In terms of cognitive features, mild to moderate IDD is typical with a characteristic pattern of cognitive strengths and weaknesses. Common behavioral features include hypersociability and high rates of psychiatric comorbidities, particularly anxiety (4-7), while medical features include but are not limited to vascular stenoses, connective tissue abnormalities, distinctive facial features, dental anomalies, and multiple endocrine abnormalities (8). Infants and toddlers with WS often experience feeding difficulties, which provides a partial explanation for failure to thrive in some young children, and slow weight gain in many. In contrast to weight findings in childhood, several small series focusing on adults with WS observed a broad BMI range (8-11). The majority of adults in these series were classified as having overweight or obesity, though a small proportion with underweight BMI were observed as well. Potential dietary or behavioral factors contributing to differences in patterns of weight change over time, however, have received essentially no scrutiny to date. Examination of the WS diet is limited to a single study (11) and so far, no work exists that compares dietary intakes among persons with WS classified by BMI or sex. Likewise, there are no data that compares persons with WS to a healthy control population or which assesses the impact of food-related behavioral problems on BMI or longitudinal weight change patterns.

The objectives of this study were to investigate BMI in adults with WS and to reveal possible factors contributing to differences in weight status classifications and weight gain in adults with WS. Our specific aims were to: 1) describe predicted intakes of selected dietary components in adults with WS and compare intakes with controls from National Health and Nutrition Examination Survey (NHANES); 2) analyze associations between BMI, predicted intake of dietary components, food-related behaviors, and anxiety in the WS study participants; and 3) for a subset of participants with WS, compare those with the highest rate of weight gain to those with stable weight over time to further explore possible factors associated with weight gain.

Methods and analyses

Subject recruitment

Individuals with WS and their families were recruited through a study announcement posted on websites of the Williams Syndrome Association (www.williams-syndrome. org), The Williams Syndrome Patient & Clinical Research Registry (www.williams-syndrome.org/registry), Rally with Mass General Brigham (https://rally.massgeneralbrigham. org/), or by hard copy invitation sent to personal patients of one of the authors (BRP). Eligibility criteria included the following: males or females between the ages 18 and 70, diagnosis of WS, and availability of a parent or guardian to participate in the consent process and provide medical history.

Written consent or assent, depending on legal guardianship status, was obtained from all WS participants in the study. An adult parent, family member, or legal guardian was also required to participate in the consenting procedures along with the person with WS regardless of legal guardianship status. This study was approved by the Mass General Brigham Human Research Committee.

Adults with WS participated in this study either in the Translational and Clinical Research Center at Massachusetts General Hospital (tCRC at MGH) between October 2018 and February 2020 or at a Williams Syndrome Association Family Convention (Baltimore MD July 11–14, 2018 or Minneapolis MN, November 7–9, 2019). Study recruitment was terminated at the start of the COVID-19 pandemic.

Study procedures

A physical examination was performed by a study physician. Height and weight measurements were obtained and these were used to calculate BMI. Participants were classified into BMI categories according to the World Health Organization criteria underweight (<18.5 kg/m²), normal (18.5–<25 kg/m²), overweight (25–<30 kg/m²) or obese (\geq 30 kg/m²) (12). Participants were also asked to sign a medical record release form authorizing study staff to obtain historic height and weight records from their primary care provider's office.

Survey data

- 1. All participants, with a family member or guardian, completed several surveys. For the purpose of this study, we developed a health and family history questionnaire, which covered topics such as past medical history, family history, weight status over time, and medication use at the time of the study.
- 2. TheDietaryScreenerQuestionnaire(DSQ), developed by the National Cancer Institute (NCI), calculates predicted mean daily intake (subsequently referred to as daily intake) using responses to 26 survey questions. Specifically, daily intakes are based on selfreported intake frequencies during the past month of multiple food and beverage item groups; frequency report options include: never, number of times last month, number of times last week, and number of times per day. Responses were analyzed using the current National Cancer Institute -recommended scoring procedures developed on data from the 2009–2010 NHANES cohort (13, 14). This scoring procedure predicts daily mean intake of selected dietary components important for the overall quality of diet, including fruits and vegetables, dairy, whole grains, fiber, calcium, added sugars, and sugar-sweetened beverages.

DSQ food frequency responses from N = 2,022 adults belonging to the 2009–2010 NHANES cohort were used as a control cohort (13). The control sample size was obtained by selecting all adult cases on whom DSQ food frequency responses were available; nonpregnant and White non-Hispanic cases were analyzed to approximate characteristics of our WS cohort.

General population daily intake recommendation and allowance values were obtained from the U.S. Department of Agriculture (USDA) and U.S. Department of Health and Human Services (HHS) Dietary Guidelines for Americans (DGA) 2020– 2025, Table 4-1 (page 96), for adults consuming a 1,800–2,400 calorie per day diet (15).

3. Food and eating behaviors in the cohort were characterized using responses to the Food-Related Problem Questionnaire (FRPQ). This is a 16-item questionnaire scored on a 7-point Likert scale from 0 (never) to 6 (always) indicating the frequency of

the behavior, with a Likert score ≥1 indicating at least some frequency of the behavior. Results are summarized in subscores of 'preoccupation with food', 'impairment of satiety', and 'composite negative behavior'. The 'composite negative behavior' can be further divided into subscores of 'takes and stores food', 'inappropriate response', and 'eats inedible items' (16). The FRPO was developed to assess eating related problem behaviors in individuals with Prader-Willi syndrome (16) but has also been used to characterize food-related problem behaviors in other genetic syndromes as Smith-Magenis syndrome and Angelman syndrome (17–19). Pica is not a feature that occurs in adults with WS, so the subscore of 'eats inedible items' was not included in the assessment. All other subscales were scored.

4. The Spence Children's Anxiety Scale-Parent version (SCAS-P), a validated parent proxy-report survey (20), was used to assesses total anxiety score. The instrument allows for scoring of six subscales: panic attack and agoraphobia, separation anxiety, physical injury fears, social phobia, obsessive compulsive, and generalized anxiety disorder. The caretaker is asked to respond to each question on a four-point Likert scale for how often each anxiety item occurs (never to always).

Weight trajectory

Data from participants who authorized access to historic heights and weights from records in their primary care provider's office were used to generate weight (kg) over time (months) trajectory plots. Individuals with two or more weight points at least 6 months apart were included in this analysis; the terminal weight point for each subject was measured during study participation.

Statistical analysis

Variables were assessed for normality using the Shapiro– Wilk test; log transformation was applied to skewed data. DSQ daily intake for each dietary component group was calculated as described in Methods section.

Dietary intake in WS and in WS versus NHANES controls

Independent samples *t*-tests examined sex differences in daily intake in the WS cohort only.

Welch's *t*-test assessed for BMI and age differences between the WS and NHANES cohorts. Linear regression analyses controlling for age assessed for daily intake differences between cohorts; this was repeated in sexseparated models also controlling for age. Bonferroni correction for the eight multiple comparisons within the whole cohort as well as the sex-specific sub cohorts, was applied to adjust the significance level (P < 0.00625).

BMI associations with dietary intake including sex differences, FRPQ food behaviors and SCAS anxiety scores

Logistic regression analyses adjusted for sex and age examined BMI category associations with DSQ daily intakes as follows: obesity versus normal weight and overweight, and underweight versus normal weight and overweight.

FRPQ subscale scores were computed for each subject. Linear regression analyses adjusted for sex and age investigated associations between BMI and FRPQ subscale scores, and between BMI and frequencies of individual food behaviors.

SCAS subscale/total scores were computed for each subject. Kruskal–Wallis tests were utilized to assess for differences between BMI categories (obesity vs. normal weight and overweight, and underweight versus normal weight and overweight) and mean SCAS subscale/total scores.

Weight trajectory modeling and analyses

A linear mixed-effects model utilizing the restricted maximum likelihood method with a diagonal covariance structure evaluated weight trajectories over time and the effects of covariates. The model included age, sex, and their interaction for fixed effects. Difference in subject starting weight was accounted for as a random effect of subject-specific intercept. An average weight change over time (kg/year) was obtained controlling for the fixed effects described here. Subject-specific slopes were obtained, and weight point specific residuals based on group mean revealed two distinct classes of weight trajectories for analysis: Group 1: 'high gain trajectory' was comprised individuals with a line of best fit slope of a magnitude in the top third of the cohort; and Group 2: 'stable weight' individuals, with a slope in the middle third of the cohort closest to zero and who did not experience two consecutive periods of gain or loss more than 2.5% of total body weight. Slopes from a third group of individuals were not used for analysis as they did not meet criteria for either Group 1 or Group 2 (see raw data for Group 1, Group 2, and Group 3 in Supplementary Fig. 1).

To explore possible factors associated with weight gain, chi-square analyses compared the two weight groups in factors such sex distribution, diagnosis of diabetes mellitus, self-reported current use of levothyroxine, or of weight-promoting medications as classified in Wharton et al. (21) and also including medroxyprogesterone. Mann–Whitney U tests were used to compare daily intake data, FRPQ (subscales and individually reported food behaviors), and SCAS scores (total and subscale) between the two weight trajectory groups. Independent samples *t*-tests investigated potential associations with birthweight.

Data were analyzed using SPSS Statistics for Windows, version 24.0 (SPSS Inc, Chicago, Ill., USA). In analyses of food behavior data STATA version 16 (StatCorp LLC, college Station, TX) was used. Figures were created using JMP Version 16 (SAS Institute Inc., Cary, NC, 1989–2021).

Results

A total of 86 adults with WS consented to participate in this study. Thirty individuals took part in the



Fig. 1. Weight trajectories (• = actual weight) in adults with WS. Group 1, panel on left, (N = 13) shows adults with high weight gain during time period of available data, while Group 2, panel on right, (N = 17) shows those with more stable weight over time. Group 1 scored higher on the FRPQ subscale "takes and stores food" but no other differences such as M:F ratio, average age, or weight promoting medications were observed. Lines = line of best fit.

tCRC at MGH, while 56 participated at a WSA Family Convention. Four participants were removed from all statistical analyses due to missing data for both dietary assessment and historic weight measurements; accordingly, the cohort available for detailed analyses consisted of 82 adults with WS. The diagnosis of WS was confirmed by genetic testing (fluorescent *in situ* hybridization [FISH] or chromosomal microarray analysis [CMA]) in 79 of the 82 individuals while, in the remaining three, the diagnosis was confirmed clinically by one of the co-authors (BRP), an experienced medical geneticist.

More female than male adults with WS took part in this study (N = 47 females; 35 males) (Table 1). There were no significant age, race, or mean BMI differences between sexes. While the male to female proportions among those in the underweight (BMI < 18.5 kg/m^2), normal (18.5– $<25 \text{ kg/m}^2$), or obese (BMI $\ge 30 \text{ kg/m}^2$) categories were comparable, there was a significantly higher percent of males with BMIs in the overweight range (BMI 25-<30 kg/m²: 37.1% M vs. 19.1% F) (Table 1).

Participants with WS as a group had a mean BMI comparable to NHANES controls (kg/m²) (27.9 \pm 8.2 WS vs. 28.7 \pm 6.9 NHANES, P = 0.936) but were, on average, younger (32.4 years ± 11.1 WS vs. 46.1 ± 14.0 NHANES, P < 0.001). Mean daily intakes for selected major food groups are shown in Table 2 for the cohort with WS and

Table 1.	Characteristics of	adult study	participants	with WS,	N = 82
----------	--------------------	-------------	--------------	----------	--------

for NHANES controls. Daily intakes of dairy and calcium were the only two significant intake differences revealed between the WS and NHANES cohorts. Specifically, daily intake of dairy in WS males $(2.6 \pm 1.1 \text{ cup equivalents WS})$ vs. 1.8 ± 0.7 cup equivalents NHANES, P < 0.001), and of calcium (1292.1 ± 292.2 mgs WS vs. 1097.3 ± 232.0 mgs NHANES, P < 0.001) were higher than in NHANES males. A similar non-significant trend was observed in females for dairy $(1.7 \pm 0.6 \text{ cup equivalents WS vs. } 1.5$ \pm 0.5 cup equivalents NHANES, *P* = 0.011) and calcium (934.0 ± 164.6 mgs WS vs. 866.6 ± 117.5 mgs NHANES, P = 0.017). A combined sex analysis retained significance for both intake categories, though the effects were attributed primarily to the excess in WS males (calcium P = 0.002; dairy P < 0.001).

Sex analyses within the WS cohort (Table 2) demonstrated that males with WS, in comparison to females with WS, had higher mean daily intakes for the following components: added sugars (P < 0.001), sugar-sweetened beverage serving equivalents per day (P = 0.050), dairy (P < 0.001), dietary fiber (P = 0.001), and calcium (P < 0.001).

Comparison to national dietary recommendations from the 2020 Dietary Guidelines for Americans (DGA) (15) showed that mean daily intakes of added sugars for both the WS and NHANES groups were greater than

Cohort characteristics	All	Males	Females	
	N = 82	N = 35	N = 47	
Years of age, median (range)	30 (18–69)	32 (19–54)	28 (18–69)	
Race white, N (%)	80 (97.6)	35 (100)	45 (95.7)	
Race other (%)	2 (2.4)	0 (0)	2 (4.3)	
Body mass index (BMI®), median (range)	27.3 (16.7–55.5)	27.6 (17.5–47.2)	25.8 (16.7–55.5)	
BMI categories, N (%)				
Underweight (BMI < 18.5)	9 (11.0)	3 (8.5)	6 (12.8)	
Normal weight (BMI 18.5–<25)	24 (29.3)	8 (22.9)	16 (34.0)	
Overweight (BMI 25-<30)	22 (26.8)*	13 (37.1)	9 (19.2)	
Obese (BMI ≥ 30)	27 (32.9)	(3 .5)	16 (34.0)	
Living situation, N (%)				
Independently	12 (14.6)	2 (5.7)	10 (21.3)	
With family	47 (57.3)	23 (65.7)	24 (51.1)	
Group home	12 (14.6)	5 (14.3)	7 (14.9)	
Other non-independently	11 (13.5)	5 (14.3)	6 (12.7)	
Zip code defined area [×] , N (%)				
Metropolitan/micropolitan	75 (91.5)	43 (89.6)	32 (94.1)	
Small town/rural	7 (8.5)	5 (10.4)	2 (5.9)	

N defines number of subjects.

*Male versus Female P < 0.05; P-values are generated by use of Kruskal–Wallis equality-of-populations rank sum tests for continuous variables and chisquare tests for categorical variables.

*BMI is calculated as [kilograms / (meters²)].

*Defined by the 2010 United States Department of Agriculture Rural-Urban Commuting Area Codes (42).

Table 2.	Predicted mean daily intake	in subjects with WS	compared with 1	NHANES controls.	Recommendations for	general pop	ulation daily
intake o	btained from Department of .	Agriculture (USDA)	and US Departs	ment of Health and	l Human Services (HH	S) Dietary G	uidelines for
America	ns (DGA) ^{&}						

Dietary component	Recommended daily intake/ allowances ^{&}	All Mean (SD)		Males Mean (SD)		Females Mean (SD)	
		WS	NHANES	WS	NHANES	WS	NHANES
Vegetables	2.5–3	2.2 (0.6)	2.3 (0.7)	2.3 (0.6)	2.4 (0.8)	2.2 (0.6)	2.2 (0.7)
Fruits	1.5–2	0.9 (0.4)	0.9 (0.5)	0.9 (0.5)	0.9 (0.5)	0.8 (0.4)	0.9 (0.4)
Dairy	3	2.1 ^{**} (0.9)	1.7 (0.6)	2.6** (1.1)	1.8 (0.7)	1.7* (0.6)	1.5 (0.5)
Total added sugar (teaspoon equivalents/day)	≤ . – 4.7	8.9∞ (9.0)	16.8 (7.1)	23.2 (11.0)	18.7 (8.1)	15.7 (5.4)	15.0 (5.4)
Sweetened beverages	NA	9.0∞ (7.5)	8.2 (6.0)	11.7 (9.6)	9.7 (6.5)	7.0 (4.8)	6.8 (5.2)
Whole grains	3–4	0.7 (0.4)	0.7 (0.4)	0.8 (0.4)	0.8 (0.4)	0.7 (0.3)	0.7 (0.4)
Dietary fiber	25–34	15.5∞ (3.0)	16.0 (3.4)	16.7 (3.0)	17.1 (3.4)	14.6 (2.7)	14.5 (3.0)
Calcium	1,000	1089.5** (289.0)	982.1 (226.6)	1292.1** (292.2)	1097.3 (232.0)	934.0* (164.6)	866.6 (117.5)

All P-values adjusted for age by linear regression.

WS versus NHANES differences that reached statistical significance are indicated by an asterisk.

*P < 0.050 (uncorrected P for multiple comparisons).

**P < 0.00625 (Bonferroni corrected P for eight multiple comparisons).

 $^{\circ}P < 0.00625$ WS within cohort male versus female difference.

[®]Based on Dietary Guidelines for Americans 2020–2025 recommendations for 1,800–2,400 calorie diet (15).

allowance range (Table 2). Notably, 51% of individuals with WS, and 46% of NHANES adults, consumed more than the daily allowance of 11.1–14.7 tsp equivalents of total added sugar per day. Mean daily intakes of fruits, whole grains, and fibers for adults with WS were lower than DGA recommendations, as were the NHANES mean intakes. 87.8% of NHANES and 82.6% of WS adults consumed less than the recommended daily intake of fruits, while 96% of NHANES and 91.9% of WS adults did so for vegetables.

WS participants with obesity (BMI \ge 30), when compared with WS participants who had either normal weight or overweight (BMI 18.5–<30), had a lower mean intake of fruits and vegetables of 0.15 cup equivalents per day, (95% confidence interval [CI] [0.01, 0.11], P = 0.049). The group of WS participants with an underweight BMI (BMI < 18.5) when compared with the groups with normal and overweight, had a reduced mean intake of fruits and vegetables of 0.44 cup equivalents per day, (95% CI [0.05, 0.82], P = 0.026); they also had a reduced intake of dietary fiber of 2.12 g/day (95% CI [0.36, 3.88], P = 0.019). No significant associations between BMI category and mean intake of dairy, added sugars or whole grains were detected.

FRPQ subscale score analyses showed the total 'preoccupation with food' sub-score was 5.87 (standard deviation [SD] 3.75) of a maximum score of 18, while the total 'impairment of satiety' sub-score was 10.33 (SD 4.15) of a maximum score of 30. Examination of specific items that comprise the 'impairment of satiety' subscale reveals a high frequency of selected food behaviors. The majority of adults with WS (70%) were reported as displaying at least some frequency of the behavior, 'if given the opportunity, eats more than a standard sized meal'. For the 'takes and stores food' subcategory, the total sub-score was 3.83 (SD 3.93) of a maximum score of 18; 87% of adults were reported displaying some frequency of helping themselves to food that they should not have, a specific item that comprises this subscale.

In a linear regression model adjusted for sex, a onepoint score increase in the FRPQ 'preoccupation with food' subscore was associated with a 0.57 unit increase in BMI, (95% CI [0.11, 1.02], P = 0.016), while a onepoint increase in the 'takes and stores food' subscore was associated with a 0.72 unit increase in BMI, (95% CI [0.30, 1.15], P = 0.001). No significant association was found between BMI and the 'impairment of satiety' sub-score.

Across the subset of participants with available historic weight data (N = 41), weight increased at an average rate of 0.08 kg/month (95% CI [0.05, 0.12], P < 0.001). For these 41 participants, mean duration of weight trajectory data was 64.43 months (range 6–228); mean age at starting weight point was 28.8 ± 9.8 years; and mean age at terminal point was 34.4 ± 10.8 years. Compared with males, the rate of weight increase was greater in females by 0.10 kg/month (95% CI [0.06, 0.13], P < 0.001).

Historic weight subset participants were classified into distinct weight trajectory groups (Fig. 1) based on mixed model analysis as follows: Group 1 (high weight gain trajectory, N = 13); Group 2 (stable weight, N = 17); and

Groups 1 and 2 had a comparable percent of males (Group 1: 38.5% male; Group 2: 33.3% male; P = 0.778). Groups 1 and 2 were also not significantly different in average age at starting and terminal measurement (starting age: Group 1: 30.7 ± 11.7 vs. Group 2: 26.8 ± 10.1, P = 0.397; terminal age: Group 1: 36.6 ± 12.4 vs. Group 2: 32.9 ± 9.6 , *P* = 0.267). Group 1 scored higher on the FRPQ subscale 'takes and stores food' with a median score of 3.00 for Group 1 and 0.00 for Group 2, P = 0.003. Additional analyses revealed no significant differences between Groups 1 and 2 including: frequency of diabetes mellitus and use of levothyroxine; frequencies of reported weight-promoting medications (P = 0.339); percentages of individuals reporting first or second-degree family members with overweight/obesity (P = 0.547); or birth weight (g) (P = 0.487). Likewise, no differences were noticed between Group 1 and Group 2 in dietary intakes, or SCAS scores (see Supplementary Table 1).

To view each subject's actual weight points included for analysis in spaghetti plot format, see Supplementary Fig. 1.

Discussion

In a cohort of adults with WS, we examined BMI cross sectionally and, in a subset, longitudinally. We observed considerable BMI variability and sought to identify contributory factors. These factors fell into two broad categories: dietary differences based on a food frequency questionnaire (DSQ) and food-related behaviors (based on FRPQ). Results from the DSQ revealed deviations from dietary recommendations for several components (15) in the WS cohort that are indicative of poor diet quality. Differences in dietary intake between BMI categories were revealed only for fruit and vegetable intake for individuals with either obesity or underweight BMI, and additionally in dietary fiber intake for individuals with underweight BMI; no other distinctions were found. Analysis of endorsement of food-related behavioral issues based on the FRPQ was more provocative and raised the possibility that problems in the behavioral domain, not previously appreciated in adults with WS, may be associated with BMI.

On average, the WS cohort, as well as the NHANES controls, consumed less than the recommended intake of fruits, vegetables, whole grains, and dietary fiber, but greater than the allowance guidelines for added sugars (13, 15). This pattern has also been observed in other adult populations with IDD (22–24).

Low intake frequency of fruits and vegetables in WS, and lower plasma carotenoids compared with other IDD groups, has previously been described in a Norwegian cohort of adolescents and adults with WS (11). Plasma carotenoids levels, a biomarker of fruits and vegetable intake, tended to be lowest in persons with WS whose BMI was <18.5 (underweight). This association of low carotenoids with low BMI is in accordance with the present data. However, all our WS subjects, regardless of BMI classification, had a lower daily intake of fruits and vegetables than recommended by the DGA (15).

The mean daily intake of dietary fiber in our WS cohort was reduced by approximately 50% of the DGA recommendation (15). A common problem, estimated to complicate the lives of half of individuals with WS, is chronic or recurring constipation that is likely related to inadequate dietary fiber intake (25). Though the food frequency intake data does not capture information on fiber supplements, not all supplements prevent constipation and furthermore, non-psyllium containing products appear not to provide the same broad range of health benefits as dietary fiber (26).

Among general population adults, numerous studies indicate that a poor-quality diet, one that is high in refined carbohydrates and low in fruits and vegetables, increases the risk of various medical problems including but not limited to obesity, diabetes mellitus, high blood pressure, and stroke (27–29). It is possible that these problems, already well-known to complicate WS, may be amplified by poor diet quality. While dietary management guidelines exist for Prader–Willi syndrome and Down syndrome, which include recommendations for reduced intake of simple carbohydrates, increased dietary fiber and fruits and vegetables (24, 30–32), no published guidelines currently exist for individuals with WS.

One final result to expand on is that the daily calcium intake in the WS males exceeded the DGA recommendation and was also significantly higher than intakes for NHANES males and WS females. Since persons with WS are at an increased risk of hypercalcemia, especially during infancy, dietary calcium intake is often moderated by caregivers/families (most commonly as limiting dairy products) (33) and this can turn into a lifelong pattern (clinical observation, BRP). Yet our results suggest this is uncommon, especially in WS males, some of whom are consuming more calcium from food sources than is recommended. If this finding is confirmed in an independent cohort, it suggests that dietary calcium intake should be routinely assessed in adults with WS before supplemental calcium is recommended.

BMI (cross-sectional data)

BMI among the WS adults in this study ranged from a low of 16.7 kg/m² to a high of 55.5 kg/m^2 (median 27.3).

The majority of the cohort (60%) was classified as having either overweight or obesity, but it is important to also point out that 11% of our study participants had a BMI in the underweight range (BMI ≤ 18.5).

Four other WS series on comparably aged adults demonstrated a similar frequency of subjects with a BMI ≥ 25 , ranging from 49 to 65% (8–11). In terms of underweight individuals, Nordstrom et al. reported 13% of the WS cohort met criteria for being underweight, which is comparable to the 11% we observed. The other three studies had 0–7% of participants with a BMI < 18.5 but were focusing primarily on the association of diabetes mellitus and WS.

BMI comparisons to other disorders with IDD is challenging due to methodologic differences and modest sample sizes. Excluding comparison to adults with Down syndrome, a group well-known to have a high prevalence of obesity, limited data suggests similar BMI distributions among adults with WS, adults with IDD of unknown etiology, and adults with autism spectrum disorder. However, being underweight may more commonly complicate the diagnosis of WS especially when compared with those with IDD of unknown etiology (34, 35). Chronically underweight individuals are at an increased risk of a variety of medical concerns including decreased bone density, impaired immunity, selected nutrient deficiencies, and fatigue among others (36). Accordingly, we recommend that further work in adults with WS be performed on the issue of underweight status in the hopes of identifying potentially remediable medical and/or psychological factors.

Food behavior

We are the first report on food-related behaviors, using the FRPQ, in adults with WS. Prevalent problematic food behaviors in WS (Table 3) included talks about food (91%), helps themselves to food they should not have (87%), responds negatively if denied food (74%), and eats more than a standard sized meal (70%). Furthermore, a stronger endorsement of food preoccupations, and of taking and storing food, are associated with an increase in BMI.

We compared FRPQ findings from the WS cohort with other adults with IDD reported in the literature, namely: 1) adults with Prader–Willi syndrome (PWS); 2) adults with Smith–Magenis syndrome (SMS) using adult data kindly extracted by Dr. Sarah Elsea from Alaimo et al. (17); and 3) adults in a generalized intellectual disability group. For the 'food preoccupation' subscale, our WS cohort scored higher on average than a generalized IDD group but lower than PWS and SMS adults (IDD = $3.8 \pm$ 2.4, WS = 5.9 ± 3.8 , PWS = 8.1 ± 2.0 , SMS = 8.6 ± 3.4) (16, 17). We also observed several more positive highly prevalent food behaviors in WS; for instance, all participants could accept an explanation if a meal is delayed, and

Individual FRPQ questions	Percent of sample with report of food behavior ¹ (%)	Average Likert scale response ¹ I–6 (SD)
Preoccupation with food		
Compares size of meal with other	59	1.80 (1.34)
Talks about food	91	3.23 (1.61)
Associates people or places and/or occasions with specific food	73	3.20 (1.42)
Impairment of satiety		
Still feels hungry after a normal sized meal	71	2.05 (1.30)
Goes without food if feeling tired, ill or upset	74	2.07 (1.34)
Shares food with others	87	2.46 (1.32)
Describes feeling full	97	4.20 (1.09)
Given the opportunity, eats more than a standard sized meal	70	2.48 (1.37)
Takes and stores food		
Helps themselves to food they should not have	87	2.75 (1.45)
Hides or hoards food	39	2.83 (1.46)
It is necessary to lock food away	23	3.11 (1.65)
Inappropriate response		
Accepts explanation if meal is delayed	100	5.12 (1.08)
Responds negatively if denied food	74	3.01 (1.68)
Become upset or anyry if a meal includes a food not expected	60	2 35 (1 41)

Table 3. Responses to individual questions in the Food-Related Problems Questionnaire (FRPQ); N = 78 subjects with WS

Including individuals responding ≥ 1 on the 7-point Likert Scale, frequency of any occurrence of behavior. Table adapted from Alaimo et al. (17). 87% can share food with others. These behaviors point toward the social aspects of meals and this a strength in persons with WS.

The food-related behavior findings in WS suggest that the development of tailored weight reduction interventions for adults with WS might need to incorporate strategies on how to tackle and avoid some of these problematic behaviors. This perspective is supported by a small study in adolescents and young adults with Down syndrome. The largest weight loss occurred in the group where parents received education in behavioral strategies in addition to individual diet and exercise plans, while parents in the comparison group only received individual diet and exercise plans (37). Behavioral interventions for food-related behaviors have also been described in Prader-Willi syndrome with success, including considerations for managing hyperphagia with cognitive behavioral therapy (38, 39). The benefit of layering behavioral interventions particularly to address preoccupation with food needs to be investigated in larger studies on persons with WS and a role for pharmacotherapy to assist in the management of food preoccupations merits consideration as well.

Weight trajectory

A dearth of information on weight trajectories in adults with IDD exist and, to our knowledge, this study is the first to investigate individual trajectories. Those in Group 1 (Fig. 1) experienced a mean rate of gain of 3.5 kg per year, and nearly all these individuals were classified as having obesity at time of study measurement (final weight point). In a search for potential causative differences between those with the highest weight gain (Group 1) and those in the stable weight group (Group 2), we did not identify any significant differences (including in male to female ratio, frequency of diabetes mellitus, use of weight promoting medications, etc.). However, an increased score for the FRPQ subscale 'takes and stores food' was noticed in Group 1 compared with Group 2 (Supplementary Table 1).

To our knowledge, only a single previous study examined weight change over time in adults with IDD (34). Ptomey and colleagues reported mean annualized percent weight change of -0.2, 1.6, and 2.2% for several hundred adults with Down syndrome, autism spectrum disorder, and IDD, respectively (34). When examining percent weight change in the present WS cohort, we find a mean 2% annualized weight change. However, due to methodologic differences in data collection, it is difficult to compare the weight change findings of Ptomey et al. (34) with ours.

In our view, further investigation of adult weight trajectories is a rich area for future study as a means to identify biological, behavioral, or environmental factors that contribute to weight gain, that might prevent weight gain, or that cause and perpetuate being underweight. And though it was not a focus of our current analyses, patterns of high weight fluctuations occurred in several of our subjects [see unclassified group (Group 3) in Supplementary Fig. 1]. Scrutinizing the medical and psychological well-being of subjects at the time of these fluctuations may reveal predictable events, which could allow preventatives to be put in place to mitigate fluctuations.

Limitations and strengths

We elected to administer the DSQ based on several strengths of the instrument, including excellent validation in a large general population cohort by comparison to 24-h recall data (14) and simple, easy to understand question content and format making it suitable for persons with WS. In addition, the DSQ allowed for comparison to a large control group (13). However, there are distinct limitations to the DSO compared with a comprehensive assessment of dietary intake, first among which is underreporting of unhealthy foods and overreporting of healthy foods, with a potentially stronger effect in persons who have overweight and obesity (40). While these limitations are inherent to any measure that collects self-reported dietary intake information, particular concerns can be raised about information from adults with WS. Reassuringly, another study of adults with WS (11) demonstrated good agreement between frequency of reported intake and measured biomarkers with dietary implications. Similarly, in this study, DSO information was captured by parent/caregiver proxy-report combined with input from the individual with WS. Since the majority (57%) of our subjects resided at home with parent/ caregivers, we believe this enhanced accuracy but appreciate this could be a limitation for individuals living independently or in a group living situation.

In terms of the FRPQ, it has been previously used in adults with PWS and SMS in multiple living situations (16, 17, 41). The recommended approach to FRPQ interpretation is to tally and compare subscale scores (16). We chose to pay particular attention to individual question responses so that we could obtain greater granularity into food-related behaviors among adults with WS. Findings from this exploratory report on food behaviors may guide future research into behavioral contributions to weight variability.

In terms of sample bias, it is possible that adults with WS and their families who have experienced issues with either low or excess body weight were more likely to take part in this study, which was described to families as focusing on factors contributing differences in body shapes and sizes. This bias does not invalidate the study's findings but does limit their generalizability to the universe of adults with WS who may not have concerns about body shape.

Lastly, analyses of weight trajectories were performed on relatively small samples sizes and were based on retrospectively collected medical records; therefore, data differed between subjects on variables such as ages for beginning and terminal weight points; duration of weight trajectory plots; time between measurements; and equipment used to obtain measurements. While a mixed model accounts for these factors, larger sample sizes and longer time intervals of data collection are needed to investigate weight trends and identify contributing dietary and non-dietary factors, including food behavior problems.

Conclusions

Considerable variation in BMI was observed among adults with WS. Mean daily intakes of fruits, vegetables, and dietary fiber in WS adults, as well as in NHANES controls, were below national dietary guideline recommended ranges. Within our WS cohort, those with a BMI in either obesity or underweight category had decreased intake of fruits and vegetables compared with normal/ overweight individuals while, additionally, individuals in the underweight category had decreased dietary fiber intake. These deviations from recommended intakes are an indicator that diet quality should be further evaluated in adults with WS.

When assessing food behaviors, increased BMI and rate of weight gain were associated with increased preoccupation with food subscores and with a higher frequency of taking and storing food. For adults with WS and obesity or rapid weight gain, attention to problematic food behaviors may be an important aspect to consider and address.

Conflict of interest and funding

H Lee: none; MA Nordstrom: none; BR Pober: Research funding to her institution from the Williams syndrome Association that supports this work; no other potential conflicts; DR Renzi: Partial effort covered by research funding to her institution from WSA; TL Stanley: Research funding to her institution from Pfizer, Inc., unrelated to this work.

We thank the funders of this work, the Williams Syndrome Association (www.williams-syndrome.org). Support for the MGH translational Clinical Research Center was provided by 1 UL1 TR002541-01.

Acknowledgments

We express gratitude to the many individuals with Williams syndrome and their families for generously donating their time and effort to take part in this study. We very much thank Dr. Sarah H. Elsea for separately examining FRPQ results on adults with Smith–Magenis syndrome previously published in Alaimo et al. Lastly, we appreciate the general assistance provided by the MGH translational Clinical Research Center staff and our research assistants Ms. Julia Pelletier and Mr. Aidan Gallagher.

References

- Rimmer JH, Yamaki K. Obesity and intellectual disability. Ment Retard Dev Disabil Res Rev 2006; 12(1): 22–7. doi: 10.1002/ MRDD.20091
- Segal M, Eliasziw M, Phillips S, Bandini L, Curtin C, Kral TVE, et al. Intellectual disability is associated with increased risk for obesity in a nationally representative sample of U.S. children. Disabil Health J 2016 Jul 1; 9(3): 392. doi: 10.1016/J. DHJO.2015.12.003
- Melville CA, Hamilton S, Hankey CR, Miller S, Boyle S. The prevalence and determinants of obesity in adults with intellectual disabilities. Obes Rev 2007 May; 8(3): 223–30. doi: 10.1111/J.1467-789X.2006.00296.X
- Royston R, Howlin P, Waite J, Oliver C. Anxiety disorders in Williams Syndrome contrasted with intellectual disability and the general population: a systematic review and meta-analysis. J Autism Dev Disord 2017 Dec 1; 47(12): 3765–77. doi: 10.1007/ S10803-016-2909-Z
- Royston R, Oliver C, Howlin P, Waite J. Anxiety characteristics in individuals with Williams syndrome. J Appl Res Intellect Disabil 2021 Jul 1; 34(4): 1098–107. doi: 10.1111/JAR.12864
- Thom RP, Pober BR, McDougle CJ. Psychopharmacology of Williams syndrome: safety, tolerability, and effectiveness. Expert Opin Drug Saf 2021; 20(3): 293–306. doi: 10.1080/ 14740338.2021.1867535
- Pitts CH, Klein-Tasman BP, Osborne JW, Mervis CB. Predictors of specific phobia in children with Williams syndrome. J Intellect Disabil Res 2016 Oct 1; 60(10): 1031–42. doi: 10.1111/jir.12327
- Pober BR. Williams-Beuren syndrome. N Engl J Med 2010 Jan 21; 362(3): 239–52. doi: 10.1056/nejmra0903074
- Lunati ME, Bedeschi MF, Resi V, Grancini V, Palmieri E, Salera S, et al. Impaired glucose metabolism in subjects with the Williams-Beuren syndrome: a five-year follow-up cohort study. PLoS One 2017 Oct 1; 12(10): e0185371. doi: 10.1371/journal. pone.0185371
- Masserini B, Bedeschi MF, Bianchi V, Scuvera G, Beck-Peccoz P, Lalatta F, et al. Prevalence of diabetes and pre-diabetes in a cohort of Italian young adults with Williams syndrome. Am J Med Genet A 2013; 161(4): 817–21. doi: 10.1002/ajmg.a.35655
- Nordstrøm M, Paus B, Andersen LF, Kolset SO. Dietary aspects related to health and obesity in Williams syndrome, Down syndrome, and Prader-Willi syndrome. Food Nutr Res 2015 Feb 3; 59: 25487. doi: 10.3402/FNR.V59.25487
- WHO Consultation on Obesity (1999: Geneva), World Health Organization. Obesity : preventing and managing the global epidemic: report of a WHO consultation. WHO Technical Report Series 894. Geneva: World Health Organization; 2000. Available from: https://apps.who.int/iris/handle/10665/42330 [cited 04 March 2022].
- 13. Centers for Disease Control and Prevention (CDC). National Center for Health Statistics (NCHS). National Health and Nutrition Examination Survey Data. Hyattsville, MD: US Department of Health and Human Services, Center for Disease Control and Prevention; 2010. Available from: https://www.cdc. gov/nchs/nhanes/index.htm [cited 04 March 2022].
- Thompson FE, Midthune D, Kahle L, Dodd KW. Development and evaluation of the National Cancer Institute's dietary screener questionnaire scoring algorithms. J Nutr 2017 Jun 1; 147(6): 1226. doi: 10.3945/jn.116.246058
- U.S. Department of Agriculture and U.S. Department of Health and Human Services. Dietary guidelines for Americans, 2020– 2025. 2020. Available from: https://www.dietaryguidelines.gov/

resources/2020-2025-dietary-guidelines-online-materials [cited 4 March 2022].

- Russell H, Oliver C. The assessment of food-related problems in individuals with Prader-Willi syndrome. Br J Clin Psychol 2003 Nov; 42(4): 379–92. doi: 10.1348/014466503322528928
- Alaimo JT, Barton LV, Mullegama SV, Wills RD, Foster RH, Elsea SH. Individuals with Smith-Magenis syndrome display profound neurodevelopmental behavioral deficiencies and exhibit food-related behaviors equivalent to Prader-Willi syndrome. Res Dev Disabil 2015 Dec 1; 47: 27–38. doi: 10.1016/J. RIDD.2015.08.011
- Welham A, Lau J, Moss J, Cullen J, Higgs S, Warren G, et al. Are Angelman and Prader-Willi syndromes more similar than we thought? Food-related behavior problems in Angelman, Cornelia de Lange, Fragile X, Prader-Willi and 1p36 deletion syndromes. Am J Med Genet A 2015 Mar 1; 167(3): 572–8. doi: 10.1002/ajmg.a.36923
- Gandhi AA, Wilson TA, Sisley S, Elsea SH, Foster RH. Relationships between food-related behaviors, obesity, and medication use in individuals with Smith-Magenis syndrome. Res Dev Disabil. 2022 Aug; 127: 104257. doi: 10.1016/j.ridd.2022.104257
- Nauta MH, Scholing A, Rapee RM, Abbott M, Spence SH, Waters A. A parent-report measure of children's anxiety: psychometric properties and comparison with child-report in a clinic and normal sample. Behav Res Ther 2004; 42(7): 813–39. doi: 10.1016/S0005-7967(03)00200-6
- Wharton S, Raiber L, Serodio KJ, Lee J, Christensen RAG. Medications that cause weight gain and alternatives in Canada: a narrative review. Diabetes Metab Syndr Obes 2018; 11: 427– 38. doi: 10.2147/dmso.S171365
- Draheim CC, Stanish HI, Williams DP, McCubbin JA. Dietary intake of adults with mental retardation who reside in community settings. Am J Ment Retard 2007 Sep; 112(5): 392–400. doi: 10.1352/0895-8017(2007)112[0392:dioawm]2.0.CO;2
- Ptomey L, Goetz J, Lee J, Donnelly J, Sullivan D. Diet quality of overweight and obese adults with intellectual and developmental disabilities as measured by the healthy eating index-2005. J Dev Phys Disabil 2013 Dec; 25(6): 625–36. doi: 10.1007/ s10882-013-9339-z
- 24. Gast DAA, de Wit GLC, van Hoof A, de Vries JHM, van Hemert B, Didden R, et al. Diet quality among people with intellectual disabilities and borderline intellectual functioning. J Appl Res Intellect Disabil 2022 Mar 1; 35(2): 488. doi: 10.1111/ jar.12958
- Morris CA, Braddock SR, Chen E, Trotter TL, Berry SA, Burke LW, et al. Health care supervision for children with Williams Syndrome. Pediatrics 2020 Feb; 145(2): e20193761. doi: 10.1542/ peds.2019-3761
- 26. McRorie JW. Evidence-based approach to fiber supplements and clinically meaningful health benefits, Part 1: what to look for and how to recommend an effective fiber therapy. Nutr Today 2015 Mar 10; 50(2): 82–9. doi: 10.1097/nt.0000000000082
- 27. Dicken SJ, Batterham RL. The role of diet quality in mediating the association between ultra-processed food intake, obesity and health-related outcomes: a review of prospective cohort studies. Nutrients 2021 Dec 22; 14(1): 23. doi: 10.3390/nu14010023
- Priano SM, Hong OS, Chen JL. Lifestyles and health-related outcomes of U.S. hospital nurses: a systematic review. Nurs Outlook 2018 Jan 1; 66(1): 66–76. doi: 10.1016/J.outlook.2017.08.013
- Clemente-Suárez VJ, Mielgo-Ayuso J, Martín-Rodríguez A, Ramos-Campo DJ, Redondo-Flórez L, Tornero-Aguilera JF.

The burden of carbohydrates in health and disease. Nutrients 2022 Sep 15; 14(18): 3809. doi: 10.3390/nu14183809

- Miller JL, Tan M. Dietary management for adolescents with Prader–Willi Syndrome. Adolesc Health Med Ther 2020 Aug; 11: 113. doi: 10.2147/ahmt.s214893
- Nutrition | National Down Syndrome Society (NDSS) [Internet]. Available from: https://ndss.org/resources/nutrition [cited 24 November 2022].
- Crinò A, Fintini D, Bocchini S, Grugni G. Obesity management in Prader–Willi syndrome: current perspectives. Diabetes Metab Syndr Obes 2018; 11: 579. doi: 10.2147/dmso.s141352
- 33. Sindhar S, Lugo M, Levin MD, Danback JR, Brink BD, Yu E, et al. Hypercalcemia in Patients with Williams-Beuren Syndrome. J Pediatr 2016 Nov; 178: 254–60.e4. doi: 10.1016/j. jpeds.2016.08.027
- 34. Ptomey LT, Walpitage DL, Mohseni M, Dreyer Gillette ML, Davis AM, Forseth B, et al. Weight status and associated comorbidities in children and adults with Down syndrome, autism spectrum disorder and intellectual and developmental disabilities. J Intellect Disabil Res 2020; 64(9): 725–37. doi: 10.1111/ jir.12767
- Hsieh K, Rimmer JH, Heller T. Obesity and associated factors in adults with intellectual disability. J Intellect Disabil Res 2014 Sep; 58(9): 851–63. doi: 10.1111/jir.12100
- Uzogara SG. Underweight, the less discussed type of unhealthy weight and its implications: a review. Am J Food Sci Nutr Res 2016; 3(5): 126–42.
- Curtin C, Bandini LG, Must A, Gleason J, Lividini K, Phillips S, et al. Parent support improves weight loss in adolescents and young adults with Down syndrome. J Pediatr 2013; 163(5): 1402–408.e1. doi: 10.1016/J.JPEDS.2013.06.081
- Carrillo-Nieto A, Landa-Ramírez E. Cognitive-behavioral therapy for hyperphagia in an adult with Prader-Willi Syndrome: pitfalls and challenges. Revista Latinoamercana de Medicina Conductual 2020; 9(2): 57–64.
- Ho AY, Dimitropoulos A. Clinical management of behavioral characteristics of Prader–Willi syndrome. Neuropsychiatr Dis Treat 2010 Apr; 6: 107. doi: 10.2147/ndt.S5560
- Thompson F, Subar A. Dietary assessment methodology. In: Nutrition in the prevention and treatment of disease. 2001; p. 3–39. doi: 10.1016/B978-012193155-1/50003-9
- 41. Ogata H, Ihara H, Gito M, Sayama M, Murakami N, Ayabe T, et al. Aberrant, autistic, and food-related behaviors in adults with Prader-Willi syndrome. The comparison between young adults and adults. Res Dev Disabil 2018 Feb 1; 73: 126–34. doi: 10.1016/J.ridd.2017.12.020
- 42. United States. US Department of Agriculture Economic Research Services. Rural-urban commuting area codes. 2019. Available from: https://www.ers.usda.gov/data-products/ rural-urban-commuting-area-codes/ [cited 24 April 2022].

*Barbara Pober

Division of Genetics Department of Pediatrics Massachusetts General Hospital 175 Cambridge St 5th floor Boston, MA 02114 USA Email: Pober:barbara@mgh.harvard.edu